Author’s response to reviews

Title: Can patient decision aids help people make good decisions about participating in clinical trials? A study protocol

Authors:

Jamie C Brehaut (jbrehaut@ohri.ca)
Alison Lott (alott@ohri.ca)
Dean A Fergusson (daferfusson@ohri.ca)
Kaveh G Shojania (kshojania@ohri.ca)
Jonathan Kimmelman (jonathan.kimmelman@mcgill.ca)
Raphael Saginur (rsaginur@ottawahospital.on.ca)

Version: 4 Date: 27 June 2008

Author’s response to reviews:

Dear Dr. Foy:

Re: 1509385366196781 – Can patient decision aids help people make good decisions about participating in clinical trials? A study protocol

Authors: Jamie C. Brehaut, Alison Lott, Dean A. Fergusson, Kaveh G. Shojania, Jonathan Kimmelman, Raphael Saginur

Thank you for your comments. The abstract in the submission system is now the same as the the abstract in the manuscript. The additional files that were uploaded were the confirmatory letter from the grant awarding body and a copy of the letter of IRB approval in principle. It had been requested that these files be submitted when re-uploading the manuscript after internal copy-editing. Please advise me if these files should have been submitted differently (as they are not related to the content of the manuscript).

Sincerely,

Jamie C. Brehaut, PhD